Author's response to reviews

Title: Drinking behaviours and blood alcohol concentration in four European drinking environments: a cross-sectional study

Authors:

Karen Hughes (k.e.hughes@ljmu.ac.uk)
Zara Quigg (z.a.quigg@ljmu.ac.uk)
Mark A Bellis (m.a.bellis@ljmu.ac.uk)
Ninette van Hasselt (nhasselt@trimbos.nl)
Amador Calafat (irefrea@irefrea.org)
Matej Kosir (info@institut-utrip.si)
Montse Juan (irefrea@cop.es)
Mariangels Duch (mduch@irefrea.org)
Lotte Voorham (lvoorham@trimbos.nl)

Version: 5 Date: 30 November 2011

Author's response to reviews:

Thank you for the opportunity to check the formatting of our manuscript. We have made the following adjustments to the manuscript:

1. In Table 3, we have moved the P values into a separate column to ensure these are clearly distinguished from the AORs. We have also capitalised the letter 'B' in the 'Lager/Beer' label and have added a full stop to the last sentence in the footnote.

2. In the title of Table 2, we have added a space between the word 'and' and '%BAC'

3. In paragraph 4 of the discussion, we have added a space between the word 'median' and '%BAC'

4. In the discussion, we have moved the word 'median' in the following sentence: "Further, individuals who were severely intoxicated were excluded to meet ethical requirements, meaning median reported alcohol consumption and BAC are likely to be underestimated." (it was previously before 'BAC').

5. In the last reference of the reference list, we have deleted a letter L that followed 'World Health Organization'.

If you require any further amendments or clarification please do not hesitate to contact us.

With best wishes

Karen Hughes